Author's response to reviews

Title: Return-to-work of sick-listed workers without an employment contract - what works?

Authors:

Sylvia J. Vermeulen (s.vermeulen@vumc.nl)
Sietske J. Tamminga (s.j.tamminga@amc.uva.nl)
Ton A.J.M. Schellart (ton.schellart@vumc.nl)
Jan Fekke Ybema (jan_fekke.ybema@tno.nl)
Han J.R. Anema (h.anema@vumc.nl)

Version: 2 Date: 9 February 2009

Author's response to reviews:

Dear Editor,

The revised manuscript contains the corrected affiliation, i.e. institutional address, of co-author Sietske J. Tamminga.

With respect to the content of the manuscript, this has not been changed.

With kind regards,

Sylvia J. Vermeulen, MD, MSc
Research Centre for Insurance Medicine AMC-UWV-VUmc
EMGO Institute for Health and Care Research – VU University Medical Centre
P.O. Box 7057
1007 MB Amsterdam
The Netherlands